Author's response to reviews

Title: Use of CNS medications and cognitive decline in the aged: a longitudinal population-based study

Authors:

Juha Puustinen (juhpuu@utu.fi)
Janne Nurminen (janurm@utu.fi)
Minna Löppönen (minlop@utu.fi)
Tero Vahlberg (tervah@utu.fi)
Raimo Isoaho (raimiso@utu.fi)
Ismo Räihä (iraiha@utu.fi)
Sirkka-Liisa Kivelä (sirkiv@utu.fi)

Version: 6 Date: 31 October 2011

Author's response to reviews:

Small revisions:

Page 3, last paragraph, first sentence: An extra character ‘’’ was deleted.

Page 4: Font size of title 'Methods and measures' has been changed to 10 (instead of 11).

Page 6: Font size of title 'Statistical analyses' has been changed to 10 (instead of 11).

Page 8: Font size of title 'Background data and the use of CNS medications' has been changed to 10 (instead of 11).

Otherwise no typographical errors were found and the manuscript is ready to be published.